## Clonal Expansion during *Staphylococcus aureus* Infection Dynamics Reveals the Effect of Antibiotic Intervention

Gareth McVicker<sup>1,2</sup>, Tomasz K. Prajsnar<sup>1,2,3</sup>, Alexander Williams<sup>1,2,3</sup>, Nelly L. Wagner<sup>1,2,4</sup>, Michael Boots<sup>5</sup>, Stephen A. Renshaw<sup>1,3,4</sup>, Simon J. Foster<sup>1,2,\*</sup>

Krebs Institute, University of Sheffield, Western Bank, Sheffield, United Kingdom, 2 Department of Molecular Biology and Biotechnology, University of Sheffield, Western Bank, Sheffield, United Kingdom, 3 MRC Centre for Developmental and Biomedical Genetics, University of Sheffield, Western Bank, Sheffield, United Kingdom, 4 Department of Infection and Immunity, University of Sheffield, Western Bank, Sheffield, United Kingdom, 5 Biosciences, University of Exeter, Cornwall Campus, Penryn, United Kingdom

## Abstract

To slow the inexorable rise of antibiotic resistance we must understand how drugs impact on pathogenesis and influence the selection of resistant clones. Staphylococcus aureus is an important human pathogen with populations of antibioticresistant bacteria in hospitals and the community. Host phagocytes play a crucial role in controlling S. aureus infection, which can lead to a population "bottleneck" whereby clonal expansion of a small fraction of the initial inoculum founds a systemic infection. Such population dynamics may have important consequences on the effect of antibiotic intervention. Low doses of antibiotics have been shown to affect in vitro growth and the generation of resistant mutants over the long term, however whether this has any in vivo relevance is unknown. In this work, the population dynamics of S. aureus pathogenesis were studied in vivo using antibiotic-resistant strains constructed in an isogenic background, coupled with systemic models of infection in both the mouse and zebrafish embryo. Murine experiments revealed unexpected and complex bacterial population kinetics arising from clonal expansion during infection in particular organs. We subsequently elucidated the effect of antibiotic intervention within the host using mixed inocula of resistant and sensitive bacteria. Subcurative tetracycline doses support the preferential expansion of resistant microorganisms, importantly unrelated to effects on growth rate or de novo resistance acquisition. This novel phenomenon is generic, occurring with methicillin-resistant S. aureus (MRSA) in the presence of  $\beta$ -lactams and with the unrelated human pathogen Pseudomonas aeruginosa. The selection of resistant clones at low antibiotic levels can result in a rapid increase in their prevalence under conditions that would previously not be thought to favor them. Our results have key implications for the design of effective treatment regimes to limit the spread of antimicrobial resistance, where inappropriate usage leading to resistance may reduce the efficacy of life-saving drugs.

Citation: McVicker G, Prajsnar TK, Williams A, Wagner NL, Boots M, et al. (2014) Clonal Expansion during *Staphylococcus aureus* Infection Dynamics Reveals the Effect of Antibiotic Intervention. PLoS Pathog 10(2): e1003959. doi:10.1371/journal.ppat.1003959

Editor: Andreas Peschel, University of Tubingen, Germany

Received November 3, 2013; Accepted January 14, 2014; Published February 27, 2014

**Copyright:** © 2014 McVicker et al. This is an open-access article distributed under the terms of the Creative Commons Attribution License, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

**Funding:** This work was funded by a Wellcome Trust Project Grant (Reference Number WT089981MA), an EU project: Predicting Antibiotic Resistance (PAR, Reference Number 241476) and from the European Community's Seventh Framework Programme [FP7-PE0PLE-2011-ITN] under grant agreement no. PITN-GA-2011-289209 for the Marie-Curie Initial Training Network FishForPharma. SAR is supported by an MRC Senior Clinical Fellowship (Reference Number: G0701932). Aquarium staff were supported by MRC Centre grant G0700091. The funders had no role in study design, data collection and analysis, decision to publish, or preparation of the manuscript.

Competing Interests: The authors have declared that no competing interests exist.

\* E-mail: s.foster@sheffield.ac.uk

## Introduction

Staphylococcus aureus is an opportunistic human pathogen that causes skin and tissue abscesses, occasionally leading to severe systemic illness and death [1]. Whilst the process of lesion formation itself is becoming better defined [2], the dynamics of a bacterial population within the host during infection are far less well understood. Whilst it is likely that infection progression and tissue tropism vary between different strains of *S. aureus* (such as those that cause osteomyelitis or keratitis [3,4]), research on zebrafish embryos and mice has indicated an important role of host phagocytes in the infection process [5–8]. This leads to an infection "bottleneck" in which a small fraction of the initial inoculum goes on to found a systemic infection. Previous studies have identified bottleneck phenomena for a range of bacterial species [9,10].

*S. aureus* is infamous for its rapid development of antibiotic resistance, which has grown increasingly more relevant with the widespread use of antimicrobials in agriculture and medicine. Alarmingly, where once it was restricted to health-care settings, drug-resistant *S. aureus* is now also found in the wider community [11,12]. Bacteria resistant to antibiotics such as methicillin and tetracycline have been shown to colonize humans in contact with antibiotic-treated livestock [13,14]. The emergence of antibiotic resistance in staphylococcal species has been the subject of much study; classically, studies and analyses have focused on the generation of resistance, persistence or tolerance due to advantageous mutations in a sensitive bacterial population challenged by

### **Author Summary**

Staphylococcus aureus is a major cause of human disease, made even more notable due to the spread of antibiotic resistance. We used a combination of animal models to study the spread of bacteria between organs during an infection and the resulting effect of antibiotic intervention. We found that S. aureus infection is highly clonal, following a "bottleneck" in which very few bacterial cells found each abscess. Despite previous in vitro research, the effect of antibiotics on S. aureus infection was poorly understood. We utilized our systemic infection models to study intervention with sub-curative antibiotic doses, such as one might encounter upon failing to complete an antibiotic course. We have shown that such doses are able to support the preferential expansion of antibioticresistant organisms during a mixed infection. This selection is due to the clonal pattern of infection, occurring despite a lack of effect on growth rate or on the spontaneous generation of resistance. Furthermore, it is generic to multiple pathogen species, including Pseudomonas aeruginosa, and antibiotic classes, such as with methicillinresistant S. aureus (MRSA) in the presence of oxacillin. Given the current debate in the field, our results have important implications for the design of properly-controlled treatment regimes.

high levels of antibiotics [15–17]. Development of resistance may come at a fitness cost [18], although *S. aureus* may be able to circumvent this cost via methods such as phenotypic switching [19]. Likewise, resistance to some drugs (including tetracycline and oxacillin) is known to be inducible [20,21].

Of current interest is the effect of sub-curative concentrations of antibiotics; that is, treatment that confers no significant improvement in the wellbeing of the host. Such low levels of antibiotics might be generated during failure to complete a treatment regime, or encountered via the hospital environment or in agriculture. The effects of a reduced antibiotic dose are the focus of several recent studies and reviews, again largely concerning the generation of mutants [22-24]. This area of research spans agriculture [13,14,25,26], medicine [27,28] and food safety [29] and examines multiple bacterial genera. For example, Pseudomonas aeruginosa has been shown to acquire antibiotic resistance more rapidly in the presence of low levels of antibiotic [30], and is known to be associated with staphylococcal infection, particularly in the lungs of cystic fibrosis patients [31]. Selection of antibiotic-resistant clones during low-level treatment in vivo, however, has not been directly examined.

In this study we use a set of isogenic, antibiotic-resistant strains to investigate the bacterial population dynamics within both zebrafish embryo and murine systemic infection models. Using these strains, we elucidate the effect of low levels of antibiotics on bacteria *in vivo*, showing that it is possible to confer an advantage to an antibiotic-resistant strain, even at levels of antibiotics that do not affect the growth rate of an antibiotic-sensitive strain or its ability to cause a lethal infection.

## Results

## Construction and characterization of antibiotic-resistant *Staphylococcus aureus* strains

In order to enhance the ability of infection dynamics studies to examine disease progression in both zebrafish embryos and mice, a set of three antibiotic-resistant *S. aureus* strains were constructed in the SH1000 background: erythromycin/lincomycin-resistant, EryR (GMSA015); kanamycin-resistant, KanR (GMSA016); and tetracycline-resistant, TetR (GMSA017). Growth of the three strains in vitro under favorable conditions exactly matched the wild type parent SH1000 (Figure S1A). Furthermore, virulence assays in the zebrafish embryo model (1500 CFU administered into the circulation at 30 hours post fertilization) resulted in the pattern of mortality one would expect from wild type bacteria (Figure S1B). Resistance cassettes were then transduced into a second strain background, NewHG, creating strains GMSA021, GMSA022 and GMSA023 (ErvR, KanR and TetR respectively). NewHG is a Newman derivative in which the gene encoding regulator SaeS has been repaired, resulting in a level of expression of *saeS* and downstream virulence factors that matches many other S. aureus strains, including the community-acquired methicillin-resistant S. aureus (MRSA) lineage USA300, as opposed to the much higher levels of expression seen in the presence of the Newman allele [32].

## Three-strain infection dynamics in the zebrafish embryo model

Previous work in the zebrafish embryo model suggested that growth of bacteria in vivo, leading to lethality, occurs after a biological "bottleneck" that often results in clonal expansion of a single strain [5]. This phenomenon was demonstrated by infecting zebrafish embryos 30 hours post fertilization with 1500 CFU of a 1:1:1 mixture of the three SH1000 strains (GMSA015-017) and enumerating the strains present in each embryo post mortem (Figure 1A). Infection with NewHG strains (GMSA021-023) gave similar results (Figure 1B). In both backgrounds, differential clonal expansion produced cases wherein either one or two strains out of three predominated (alongside cases wherein all three strains grew equally). There was no preference for the growth of any particular strain over its competitors, with all three antibiotic resistance cassettes proving equally capable of being selected (p>0.1), in agreement with previously published results [5]. Furthermore, we isolated clones that had been selected in vivo and, after mixing with "naïve" bacteria, reintroduced them into the zebrafish model. We found that there was no increase in the selection of these clones relative to their naïve counterparts over three successive passages, suggesting a lack of adaptive mutations, as expected [5]. Therefore in vivo clonal population expansion of the three strains is likely to be entirely stochastic and, hence, endpoint enumeration of these



**Figure 1. The stochastic distribution of bacterial strains** *post mortem* **in zebrafish embryos.** Embryos were infected with a 1:1:1 mixture of three antibiotic resistance-marked, but otherwise isogenic, strains. Each pie chart represents a single embryo infected with *S. aureus* (A) SH1000 variants or (B) NewHG variants. Total bacterial load in each case was approximately 10<sup>6</sup> CFU. doi:10.1371/journal.ppat.1003959.g001

strains is a suitable and reliable method to investigate  $in \ vivo$  dynamics.

# A mixed, three-strain infection reveals complex pathogen dynamics within the murine host

It is known that intravenous S. aureus infection results in kidney abscesses [2,33] and that these likely originate from a small number of founding bacteria [5]. Modeling of infection dynamics has been performed in the murine model using other organisms, including Salmonella enterica [9], Bacillus anthracis [10] and Yersinia pseudotuberculosis [34]. Herein, a three-strain inoculum was used to investigate the dynamics of systemic staphylococcal disease. Female BALB/c mice were infected intravenously at 7-8 weeks of age with a mixed inoculum of NewHG strains (GMSA021-023) at a 1:1:1 ratio totaling  $1 \times 10^7$  CFU. At 30 minutes and then at 2, 6, 12, 24, 48 and 72 hours post infection, mice were sacrificed and their visceral organs (kidneys, liver, spleen, heart and lungs) were harvested for bacterial enumeration by homogenization, serial dilution and plating on selective media (n = 8-10 mice per timepoint). Bacterial numbers in the blood were found to be consistently low, approaching the limit of detection at all timepoints, and are therefore not included in the analysis. Pilot data indicated that the bacteria were unlikely to be found at other sites (including brain, thigh muscle/bones and the site of infection, i.e. tail).

A large fraction of the bacterial inoculum was restricted to the liver and spleen for several hours post infection. Neither extreme clonality nor an overall increase in bacterial numbers were observed at this stage, indicating that little bacterial growth was occurring, or that there was an equilibrium between bacterial growth and killing by the host (Figure 2). As the infection progressed, bacteria were found transiently in the heart (peaking at 12 hours post infection) and then in increasingly large numbers in the kidneys (peaking at 2-3 days post infection) (Figure 2A). Bacteria in the kidneys showed a high degree of clonality, in most cases representing only one or two strains out of three. Occasionally, highly-clonal bacterial populations were found in the liver at later time-points. This dramatic increase in clonality over time is shown in Figure 2B. Data for individual organs at each time-point are given in supplementary Figure S2. One mouse (labelled "\*" in Figure S2) was sacrificed for humane reasons due to ill health at 42 hours post infection and was excluded from the general analysis.

# Sub-curative concentrations of tetracycline alter the dynamics of a staphylococcal infection in zebrafish

Clonal expansion of bacteria has important implications during mixed-strain infections, especially when a drug-resistant mutant is present among a population of sensitive organisms, and the zebrafish infection model allowed us to investigate this process. The antibiotic dose required to produce a significant reduction in mortality in the zebrafish embryo model can, in some cases, approach fifty times the *in vitro* MIC. It was hypothesized that a low antibiotic dose, i.e. one that produces no response in fish infected with a drug-sensitive strain, might still offer a preferential advantage to drug-resistant bacteria, as has been observed *in vitro* [23].

In order to define an appropriate drug concentration, tetracycline response experiments were conducted, whereupon embryos infected with 1500 CFU SH1000 EryR (GMSA015) were immersed in sterile E3 medium containing a range of antibiotic doses for the duration of the experiment (Figure S3A). While 50  $\mu$ g/ml tetracycline produced a highly significant, curative response (p = 0.0003), 10 µg/ml tetracycline showed a trend towards curing (p = 0.0753) and 5 µg/ml was entirely sub-curative (p = 0.2181). We therefore chose 2.5 µg/ml tetracycline for subsequent experiments. Embryos treated with sub-curative 2.5  $\mu$ g/ml tetracycline contained approximately 10<sup>6</sup> CFU of EryR bacteria upon death (Figure S3B), demonstrating no reduction in terminal bacterial load at this dose. This control indicates that any observed skewing of the population in favor of a resistant strain at this antibiotic concentration would not result from failure of the sensitive strain to grow to normal levels in vivo. Lastly, bacterial growth kinetics were assayed in vivo using either the EryR or TetR strain injected alone into zebrafish embryos and either left untreated or treated with 2.5 µg/ml tetracycline. Neither strain showed significantly different growth kinetics in vivo in the presence of tetracycline (p>0.9) (Figure S4), indicating that the skewing effect is not simply due to a change in growth rate within the host organism. Growth rate controls were not performed in vitro because the antibiotic concentrations involved would not correspond well with in vivo work. Experiments performed by others show that low antibiotic doses can indeed select for resistant mutants in vitro over the long term [23], but that the effect requires a far greater number of generations than exhibited by bacterial growth in our in vivo experiments.

To ascertain whether a sub-curative dose of antibiotic was indeed able to influence bacterial population dynamics, embryos were infected with a total of 1500 CFU SH1000 EryR and TetR (GMSA015 and GMSA017 mixed 1:1), and treated with subcurative 2.5 µg/ml tetracycline. Treated and untreated embryos showed no difference in mortality (Figure 3A, p = 0.7412) or in total bacterial numbers per embryo upon death (Figure 3B, p = 0.1452). There were, however, significant differences in the ratios between the strains isolated from treated and untreated embryos (Figure 3C, p = 0.0143). Lowering the tetracycline dose to 1 µg/ml abolished this effect (Figure S3C, p = 0.1008), implying a narrow window (approximately four-fold) in which the antibiotic dose is low enough to be statistically sub-curing but high enough to influence population dynamics. Bacteria recovered from these experiments were invariably found to be resistant to a single antibiotic alone, therefore the change in output ratio is not due to the spontaneous generation of resistance among the sensitive population (as bacteria that had acquired resistance to tetracycline in this way would remain erythromycin-resistant). Phagocyte depletion of infected zebrafish, which has been previously shown to prevent stochastic population variation [5], resulted in untreated and treated groups that were statistically similar even after treatment with a higher dose of 10 µg/ml tetracycline (Figure 3D; p = 0.4855), indicating that phagocytes play a role in the antibiotic skewing phenomenon, and again supporting the hypothesis that phagocytes are an important host niche during S. aureus infection.

## *In vivo* selection of resistant clones occurs among other strains, genera and antibiotic classes

To discover whether the phenomenon is consistent among other *S. aureus* strain backgrounds, the same experiments were conducted using NewHG EryR and TetR (GMSA021 and GMSA023). 2.5  $\mu$ g/ml tetracycline was again shown in pilot experiments to be non-curative in fish infected with the sensitive strain alone (Figure S5A, p = 0.5443). Results of the NewHG mixed population experiment (1500 CFU, 1:1) were consistent with those obtained for SH1000, showing similar embryo mortality between the treated and untreated groups (Figure S5B, p = 0.4575), similar total CFU recovered from both groups (Figure S5C, p = 0.4068) and a statistically significant difference between the strain ratios



**Figure 2. The bacterial strain distribution in mouse organs at various time-points post infection.** Mice were infected with a 1:1:1 mixture of three marked *S. aureus* NewHG strains (n = 8–10 mice per time-point). (A) The change in total CFU load over time in each organ (BD: below limit of detection). (B) The number of organs at each time-point that show either single-strain dominance (one strain  $\geq 100 \times$  the other two), double-strain dominance (two strains  $\geq 100 \times$  the third), spread dominance (highest and lowest strains differ by  $\geq 100 \times$ , with the middle strain within  $100 \times$  of both) or no dominance (all strains are within  $100 \times$  of one another). doi:10.1371/journal.ppat.1003959.g002

(Figure 4A, p = 0.0296). As expected, this was abolished in phagocyte-depleted embryos at 10 µg/ml tetracycline (Figure S5D, p = 0.2956). Therefore the preferential selection of antibiotic-resistant bacteria at low levels of antibiotic is not limited to a single *S. aureus* strain.

The Gram-negative pathogen *Pseudomonas aeruginosa* is commonly isolated alongside *S. aureus* in the lungs of patients with cystic fibrosis [31] and is able to modulate *S. aureus* growth and virulence factor expression during polymicrobial wound infection [35]. To explore the possibility that extreme skewing of strain ratios extends to genera of bacteria other than *Staphylococcus*, two drug-resistant *P. aeruginosa* PAO1-L derivatives, gentamicin-resistant GmR (GMPA001) and tetracycline-resistant TetR (GMPA002), were constructed. As expected from previous work by others [36,37],



**Figure 3. The effect of a sub-curative antibiotic dose on zebrafish embryos infected with** *5. aureus* **SH1000.** Embryos were infected with a 1:1 mixture of SH1000 EryR:TetR bacteria and treated with 2.5  $\mu$ g/ml or 10  $\mu$ g/ml tetracycline as indicated. (A) Zebrafish mortality curve (n = 60–65 per group). (B) Total terminal bacterial load per embryo. (C) Terminal EryR/TetR strain ratio per embryo. (D) Terminal EryR/TetR strain ratio per *pu*.1 morphant (phagocyte-depleted embryo). Solid lines indicate mean (B) and median (C, D) values. doi:10.1371/journal.ppat.1003959.g003

live PAO1 (but not heat-killed PAO1) is lethal to zebrafish embryos when injected into the circulation, even at infectious doses as low as 100 CFU (Figure S6A). P. aeruginosa has a naturally high intrinsic resistance to tetracycline and was therefore unaffected by all attempts to cure the infection (Figure S6B) using drug doses that were not harmful to the embryo ( $\leq 100 \ \mu g/ml$ ). In mixed population experiments with 50 µg/ml tetracycline (200 CFU, 1:1), it was again observed that a lack of significant effect on embryo mortality (Figure S6C, p = 0.9449) or terminal CFU load (Figure S6D, p = 0.4474) did not prevent the selection phenomenon from occurring (Figure 4B, p = 0.0169). It is noteworthy, however, that P. aeruginosa does not reach a specific bacterial load before embryo mortality (unlike S. aureus, which consistently reaches approximately 10<sup>6</sup> CFU). Furthermore, in contrast to the S. aureus strains tested above, phagocyte depletion of the host does not abolish the selection phenomenon in P. aeruginosa (Figure S6E, p < 0.0001). Thus, despite the difference between S. aureus and P. aeruginosa population kinetics (and therefore infection dynamics) in vivo, the two species exhibit the same phenomenon with regards to selection at sub-curative antibiotic doses.

It was hypothesized that the effect of sub-curative antibiotic doses on strain ratios might not be restricted to tetracycline. To test this, clinical MRSA isolate BH1CC and its methicillinsensitive, tetracycline-resistant isogenic partner [38] were used.

Experiments were carried out in an identical manner to tetracycline (i.e. immersion of embryos in antibiotic). It was found that 32  $\mu$ g/ml oxacillin had no effect on the mortality of zebrafish infected with 1500 CFU of either the sensitive strain alone (Figure S7A, p = 0.5390) or a 1:1 mixture of sensitive and resistant strains (Figure S7B, p = 0.2087). Although the total bacterial load upon death was marginally but significantly decreased using this dose of oxacillin (Figure S7C, p < 0.0001), this may be explained by the bactericidal action of oxacillin versus the bacteriostatic nature of tetracycline used in previous experiments. Nonetheless, such a decrease had no effect on the disease state or mortality of the infected host. This sub-curative dose of oxacillin produced a highly significant skew towards the resistant strain (Figure 4C, p = 0.0003), indicating that the phenomenon is not restricted to tetracycline and is likely to be an important factor in the clinical treatment of mixed MRSA-MSSA infections. Curiously, the skewing effect was also evident to a minor but significant degree in phagocyte-depleted embryos (Figure S7D, p = 0.0018) treated with the same oxacillin dose, suggesting both phagocyte dependent and independent effects. Selection was not significant at 16  $\mu g/ml$ oxacillin, suggesting a 2-fold concentration range for the effect (Figure S7E, p = 0.0964).

Interestingly, antibiotics have been proposed to share a common killing mechanism involving accumulation of reactive



**Figure 4. The effect of sub-curative antibiotic doses on zebrafish embryos infected with a variety of pathogens.** Embryos were infected with a 1:1 mixture of (A) *S. aureus* NewHG EryR:TetR bacteria (treated with 2.5 µg/ml tetracycline), (B) *P. aeruginosa* PAO1-L GmR:TetR bacteria (treated with 50 µg/ml tetracycline), (C) *S. aureus* BH1CC OxS:OxR bacteria (treated with 32 µg/ml oxacillin) or (D) *S. aureus* SH1000 EryR:TetR bacteria (treated with 100 µg/ml erythromycin). Terminal sensitive/resistant strain ratio per embryo is shown in each case. Solid lines indicate median values.

doi:10.1371/journal.ppat.1003959.g004

oxygen species (ROS) within bacterial cells, and subsequent death from oxidative stress, irrespective of the antibiotic target or direct mechanism of action [39]. However, recent reports have questioned the role of ROS in the activity of bactericidal antibiotics [40,41]. Therefore, the important components of staphylococcal oxidative stress resistance AhpC and KatA [42] were tested for their role in resistant clone selection *in vivo* (Figure S7F). Since the virulence of the *katA ahpC* double mutant (strain KC043) was not affected by oxacillin relative to its SH1000 parent at all concentrations tested (p>0.74), it is unlikely that ROS play a role in antibiotic skewing in the *in vivo* infection model.

Erythromycin is another medically relevant drug which our strain constructs enabled us to test. In the zebrafish embryo model, however, erythromycin proved unable to affect the outcome of infection by immersion alone. Instead, the antibiotic dose was introduced by microinjection not more than two hours after bacterial infection (untreated controls were instead injected with PBS). Results showed no significant difference between treated and untreated groups when comparing mortality during a single-strain infection (Figure S8A, EryR p = 0.0804, TetR p = 0.9964) or during a mixed infection (Figure S8B, p = 0.2695), nor was there a significant difference in terminal bacterial load (Figure S8C, p = 0.4438). Yet, again, there was a significant difference in terminal strain ratios (Figure 4D, p = 0.0022). Similarly to

tetracycline, no cross-resistance was observed in the strains recovered from embryos *post mortem*. Therefore, preferential selection of resistant strains (not associated with the spontaneous generation of resistant mutants) results from treatment with subcurative doses of multiple antibiotic classes.

## *In vivo* selection of resistant clones occurs in the mammalian host

In order to extend our findings to mammalian infection, antibiotic skewing experiments were performed using tetracycline in the murine model. Briefly, mixed-strain infections were carried out as for the population dynamics experiments above, and mice were treated with low doses of tetracycline in their drinking water. Mice were sacrificed two days post infection, when the majority of bacteria were expected to be found in clonal abscesses within organs (i.e. kidneys, liver and spleen), which were assayed for bacterial load.

In preliminary experiments, the sub-curative tetracycline dose varied between animal groups. Two doses (0.1 mg/ml and 0.2 mg/ml) were therefore used in the final experiment alongside an untreated control (Figure 5). There was no significant difference in total CFU per mouse for either dose compared to the control (p = 0.1965 and p = 0.4194 respectively). Pooled data (i.e. the sum total of bacteria of each strain per mouse) showed a trend towards



Figure 5. The effect of sub-curative antibiotic doses on mice infected with *S. aureus* NewHG. Mice were infected intravenously with a 1:1 mixture of NewHG EryR:TetR bacteria and were treated by replacing their drinking water with the indicated concentrations of tetracycline prior to sacrifice two days later. (A) Total CFU per mouse at experimental endpoint. (B) Total EryR/TetR strain ratio per mouse at experimental endpoint. Solid lines indicate mean (A) and median (B) values. doi:10.1371/journal.ppat.1003959.g005

the resistant subpopulation. The variance in the untreated group compared to treated groups makes comparison by Mann-Whitney test highly misleading in this model without using hundreds of animals per group (an amount that would be impossible for both practical and ethical reasons); therefore, a binomial distribution was used to analyze these data. The underlying assumption of this conservative test is that in an untreated mouse, there is a 50% chance of skewing towards either strain (which is indeed what we observed; 5/10 towards TetR; p = 0.246). The comparison showed that whilst 0.1 mg/ml did not give a significantly non-stochastic result (6/10 TetR, p = 0.205), 0.2 mg/ml did (8/9 TetR and one mouse not skewed, p = 0.0176). Therefore a sub-curative antibiotic dose is able to cause preferential selection of resistant microorganisms during mammalian infection. Furthermore, as expected from our murine infection kinetics experiment, selection at this low antibiotic concentration was a result of growth in the kidneys (p = 0.0349) rather than liver (p = 0.2188) or spleen (p = 0.2344) (Figure S9).

## Discussion

Although *Staphylococcus aureus* is an important and increasingly antibiotic-resistant human pathogen, little is known about its

disease progression or how antibiotic levels affect population dynamics. This is particularly pertinent for the understanding of *S. aureus* behavior in a hospital environment, in agriculture or even in the community at large, given the failure of some patients to complete prescribed antibiotic courses.

Construction of a set of isogenic strains that produce clonal lesions in the host demonstrated stochastic strain variance during zebrafish embryo infection. In the mouse it was shown that kidney abscesses form in the later stages of infection and each likely originates from a single bacterial founder. Interestingly, most of the initial inoculum is localized to the liver and spleen during the early stages of infection (up to approximately one day post intravenous administration). It is unknown whether bacterial cells founding renal abscesses were physically associated with the kidneys during this critical period, or whether individual bacteria "seed" the kidneys after travelling from another organ such as the liver, spleen, lungs, heart or blood at a later stage. Observation of a mouse that showed a dramatic, systemic infection of just two strains strongly implies that spread to secondary organs can occur after an initial period of clonal expansion. The variable speed at which growth occurs in the kidneys may be indicative of lesion founders "seeding" the kidneys at different times, especially in the case where two or more strains reach substantial but unequal numbers, given that no strain has a competitive advantage in terms of growth rate. In a peritoneal infection model, Rauch and colleagues reported parenchymal kidney abscesses that were identical to those observed post-intravenous infection, and that peritoneum-associated kidney surface abscesses were not associated with parenchymal abscesses [43]. Despite their role in controlling infection [44], host phagocytes may form the reservoir or delivery vehicle for S. aureus infection [5-8].

Although *S. aureus* colonization is not directly correlated with mortality [45], the presence of a mixed population of resistant and sensitive bacteria in the host provides a convenient "head start" for resistant bacteria to outgrow the sensitive strain and reach transmissible levels, should the patient become exposed to antibiotics. In the absence of treatment, however, resistance to antimicrobials has typically been thought to impose a fitness cost that enables the parent strain to outcompete antibiotic-resistant strains [18]. We have shown that this is not always the case, as even in the absence of apparent selective pressure, clonal expansion ensures that a pool of resistant bacteria can grow to dominant levels.

The use of antibiotics is the primary driving force behind the development of resistance [24]. It has been shown in S. aureus that vancomycin treatment is able to greatly increase a pool of resistant organisms inside a host [46] and that de novo resistance to tetracycline can arise when the drug is given orally at high levels [47]. Here, a combination of zebrafish embryo and murine systemic infection models were used to examine the role of subcurative concentrations of antibiotics against multiple strains of S. aureus and a Gram-negative pathogen, P. aeruginosa, in vivo. A level of drug so low that it does not alter host mortality or bacterial load (i.e. a dose that is sub-curative) is able to produce a statistically significant skew in strain ratios towards a pre-existing resistant subpopulation. In *Escherichia coli*, low levels of antibiotics such as  $\beta$ lactams promote stepwise mutations leading to resistance [22] and in Pseudomonas aeruginosa, sub-lethal concentrations of several antibiotics increase mutational frequency [30]. Sub-MIC doses of antibiotics have been demonstrated not only to contribute to the generation of resistant mutants, but also to allow a resistant organism to outcompete its sensitive competitors in long-term in vitro experiments [23].

Critically, our result is not dependent upon development of resistance in the sensitive subpopulation. Furthermore, selection of resistant clones is not simply due to retardation of growth of the sensitive bacteria caused by the antibiotic, as phagocyte-depleted zebrafish embryos (wherein bacteria grow exponentially from the outset of infection [48]) do not show a significant skew in favor of the resistant strain, nor did in vivo growth kinetics experiments reveal a significant difference. Since phagocyte-depleted embryos succumb to even a small bacterial dose within 18 hours [48], the restricted growth of the pathogen within the first 24 hours during our experiments suggests that all bacterial cells are captured by phagocytes immediately upon infection, therefore preferential phagocytosis is unlikely to be a factor. Instead, it is likely that the antibiotic confers an advantage to the resistant strain over the sensitive strain that allows it to better colonize the pre-expansion "niche" or somehow exploit that "niche" more effectively. This phenomenon may be limited to organisms whose mode of pathogenicity relies upon the stochastic selection of clones during infection, but is not restricted to a single species or class of antibiotic. It may be that such organisms are sensitive to low-dose treatment at a particular stage of their infectious cycle, in a nichedependent fashion. Since the role of reactive oxygen species (ROS) in the action of antibiotics is currently under debate [39-41] and could provide an explanation for the skewing phenomenon, we examined the effect of ROS during treatment using an S. aureus strain (ahpC katA) defective in oxidative stress resistance [42]. Despite extreme sensitivity to oxidative stress the ahpC katA strain was no more susceptible to tetracycline in vivo than its isogenic parent, suggesting that ROS may not play a major role in a generalized antibiotic effect. It is well known that the bacterial transcriptome/proteome can fluctuate when exposed to subcurative antibiotic doses [49-52], and this is an area of study that can be pursued in future work.

Environmental concentrations of antibiotics are commonly found in the ng/L to µg/L range except in extreme cases [53,54], several orders of magnitude lower than the concentrations tested in our in vivo work. We have focused, therefore, on medical treatment of infection. The global spread of antibiotic resistance is a serious threat to human health that must be acted on, as it is inevitable that high-dose antimicrobial chemotherapy increases the selective pressure on the target organism. This has led to an argument that aggressive treatment regimes, such as those currently prescribed, should be reconsidered and that alternative patterns of treatment might be indicated [55]. As discussed above, however, it is becoming accepted that lower drug doses promote stepwise development of resistance and that aggressive drug use dramatically reduces the potential pool of clones from which resistance can develop. Given this information and our data showing that sub-curative antibiotic doses are still able to select for pre-existing resistant organisms, we suggest that removal of the sensitive bacteria before they can develop resistance is still the best strategy for control of microbial disease. Although we have not determined the concentrations of antibiotics active in host tissues, we note that the elimination half-lives of the antibiotics tested in this study vary considerably, from 30 minutes (oxacillin) to 6-12 hours (tetracycline) [56]. Coupled with our evidence that the sub-curative but selective dose range is two- to four-fold, we hypothesize that bacteria are likely to be in contact with a relevant drug concentration for ample time for selection to occur during treatment. Furthermore, many antibiotics including oxacillin and tetracycline share a renal route of elimination [56], which concurs with our murine experiments suggesting that the kidneys are important for the antibiotic effects discussed. The exposure time would be greatly amplified by "pulsed" antibiotic therapy. The effect of sub-curative antibiotics may seem modest, however, we note that the evolution of any organism does indeed occur as a result of modest competitive advantages. We believe that our results are indicative of a growing trend in the response of bacterial pathogens to antibiotics, and conclude that carefully prescribed, high-dose antimicrobial chemotherapy remains preferable over the alternatives at this time.

#### **Materials and Methods**

#### Ethics statement

Animal work (both mice and zebrafish) was carried out according to guidelines and legislation set out in UK law in the Animals (Scientific Procedures) Act 1986, under Project Licenses PPL 40/ 3123, PPL 40/3699 and PPL 40/3574. Ethical approval was granted by the University of Sheffield Local Ethical Review Panel.

#### Bacterial strains and growth conditions

Staphylococcus aureus strains (Table S1) were grown using brain heart infusion (BHI) liquid or solid medium (Oxoid) at 37°C, supplemented with the following antibiotics where appropriate: kanamycin 50 µg/ml, tetracycline 5 µg/ml or erythromycin 5 µg/ml plus lincomycin 25 µg/ml (Sigma-Aldrich). *Pseudomonas aeruginosa* strains (Table S1) were grown using Luria-Bertani (LB) liquid or solid medium (Oxoid) at 37°C, supplemented with the following antibiotics where appropriate: tetracycline 125 µg/ml or gentamicin 20 µg/ml (Sigma-Aldrich).

## Construction of antibiotic-resistant S. aureus strains

The suicide vector pMUTIN4 [57] was used to integrate various antibiotic resistance cassettes downstream of the lysA gene (which encodes the terminal enzyme in the lysine biosynthetic pathway) in S. aureus. This provided a convenient method of screening to ensure that clones retained a wild type phenotype; undesired disruptions of *lysA* would result in lysine auxotrophy. Growth of clones in chemically defined minimal medium lacking lysine showed that the wild type *lysA* gene remained intact. After construction by standard PCR and restriction/ligation methods, plasmids were introduced into S. aureus RN4220 by electroporation, whereupon they integrated into the chromosome via homologous recombination. The resulting resistance markers were then transferred into other S. aureus strains as required by  $\Phi$ 11 transduction. pMUTIN4 provided erythromycin/lincomycin resistance (EryR), whereas pAISH1 [58] was used to integrate tetracycline resistance (TetR) and pGM072 (pMUTIN4 in which erythromycin resistance was replaced by the resistance cassette from pGL433 [59]) was used to integrate kanamycin resistance (KanR). BH1CC and its derivative [38] were kindly provided by James O'Gara (University College Dublin).

#### Construction of antibiotic-resistant *P. aeruginosa* strains

*P. aeruginosa* PAO1-L derivatives were constructed by integration of mini-Tn7 to a neutral locus according to published protocols [60]. Plasmids containing mini-Tn7 marked with either a gentamicin or tetracycline resistance cassette (GmR or TetR respectively) were kindly provided, along with invaluable assistance, from Stephan Heeb (University of Nottingham).

### Zebrafish maintenance and microinjection

London wild-type (LWT) zebrafish embryos (bred in the MRC CDBG aquarium facilities at the University of Sheffield; see Ethics Statement) were used for all experiments and were incubated in E3 medium at 28°C according to standard protocols [61]. In order to obtain phagocyte-depleted embryos, morpholino-modified

antisense oligomers against pu.1 [62] were injected using a method described previously [48]. Anaesthetized embryos at 30 hours post fertilization were embedded in 3% w/v methylcellulose and injected individually using microcapillary pipettes filled with bacterial suspension of known concentration into the circulation, as previously described [48]. Following infection, embryos were kept individually in 100 µl E3 medium (with or without experimental antibiotics), observed frequently up to 92 hours post infection, dead embryos removed and numbers recorded at each time point.

## Intravenous mouse injections

Female BALB/c mice were purchased from Harlan (UK) and maintained at the University of Sheffield using standard husbandry procedures. The 7–8 week old mice were inoculated in the tail vein with 100  $\mu$ l of *S. aureus* suspension in endotoxin-free PBS (Sigma) corresponding to  $1 \times 10^7$  CFU per mouse. Viable bacteria in the inoculum were plated on BHI (plus appropriate antibiotics) after serial decimal dilution to confirm the accuracy of the bacterial dose. Mice were monitored and sacrificed at various time-points according to experimental design.

#### Determination of in vivo bacterial load

In order to recover bacteria from host tissues, whole zebrafish embryos or mouse organs were individually homogenized in a suitable volume of PBS using the PreCellys 24-Dual (Peqlab). Homogenates were serially diluted in PBS and plated on BHI (*S. aureus*) or LB (*P. aeruginosa*) agar supplemented with appropriate antibiotics to determine bacterial numbers.

#### Statistical analysis

Survival experiments were evaluated using the Kaplan-Meier method. Comparisons between curves were performed using the log rank test. For comparisons between two CFU groups, a two-tailed, unpaired Student's *t*-test was used. For comparisons of strain ratios between two groups (e.g. treated and non-treated) in zebrafish embryos, a (non-parametric) Mann-Whitney *U* test was used. Analysis was performed using Prism version 6.0 (GraphPad) and statistical significance was assumed at p<0.05. In addition, analyses via generalized linear models, linear models on transformed data and Kruskall Wallace tests were performed and provided the same insights. Strain ratios in mice were analyzed by binomial distribution, as explained in Results. Figures show significance to 1 s.f. (p<0.05) or NS (p≥0.05), and indicate either mean (CFU comparison) or median (ratio comparison) values as appropriate.

#### Supporting Information

Figure S1 Comparison of fitness of antibiotic resistance-tagged SH1000 strains. (A) Growth in aerated BHI medium at  $37^{\circ}$ C. (B) Mortality of zebrafish embryos infected with each strain (n = 20–25 per group). (TIF)

Figure S2 The pattern of clonality observed during a systemic murine infection. Pie charts show the distribution of three strains in each organ, for each mouse at each time point. Numbers inside rings indicate  $\log_{10}(CFU)$  in each organ (i.e. the total bacterial load). Where total load was <100 CFU, strain distribution is not given, as presenting data this close to the detection limit would be non-representative of true strain dominance. The mouse marked \* was culled 6 hours earlier than the other mice in that group due to ill health and was not included in general analyses throughout the paper. (TIF)

Figure S3 The effect of tetracycline on zebrafish embryos infected with *S. aureus* SH1000 strains. (A) Mortality of zebrafish infected with SH1000 EryR alone and treated with a range of tetracycline doses (n = 25-30 per group). (B) Terminal EryR bacterial load per EryR-infected embryo, treated with 2.5 µg/ml tetracycline. (C) Terminal EryR/TetR strain ratio per embryo infected with a 1:1 mixture of SH1000 EryR:TetR bacteria and treated with 1 µg/ml tetracycline. Solid lines indicate median values. (TIF)

Figure S4 The effect of tetracycline on growth of *S. aureus* SH1000 *in vivo*. Graphs show growth kinetics of (A) EryR and (B) TetR. Zebrafish embryos were either left untreated (circles) or treated with 2.5  $\mu$ g/ml tetracycline (squares). Bacterial CFU loads in living fish (n = 5 per group per time-point) at 1 and 7 hours post infection and every 12 hours thereafter, were determined (white). Bacterial loads in any dead fish at each time-point were also determined (black).

(TIF)

Figure S5 The effect of tetracycline on zebrafish embryos infected with *S. aureus* NewHG strains. (A) Mortality of zebrafish infected with NewHG EryR alone, treated with a range of tetracycline doses (n = 30–35 per group). (B) Mortality of zebrafish infected with a 1:1 mixture of NewHG EryR:TetR, treated with 2.5 µg/ml tetracycline (n = 70–75 per group). (C) Total terminal CFU load per embryo infected with a 1:1 mixture of NewHG EryR:TetR, treated with 2.5 µg/ml tetracycline. (D) Terminal EryR/TetR strain ratio per *pu*.1 morphant (phagocyte-depleted embryo), treated with 10 µg/ml tetracycline. Solid lines indicate mean (C) and median (D) values.

(TIF)

Figure S6 The effect of tetracycline on zebrafish embryos infected with *P. aeruginosa* PAO1 strains. (A) Mortality of zebrafish inoculated with either live bacteria, heatkilled bacteria or sterile PBS (n = 30-35 per group). (B) Mortality of zebrafish infected with PAO1 GmR alone, treated with a range of tetracycline doses (n = 55-60 per group). (C) Mortality of zebrafish infected with a 1:1 mixture of PAO1 GmR:TetR, treated with 50 µg/ml tetracycline (n = 65-70 per group). (D) Total terminal CFU load per embryo infected with a 1:1 mixture of PAO1-L GmR:TetR, treated with 50 µg/ml tetracycline. (E) Terminal GmR/TetR strain ratio per *pu*.1 morphant (phagocytedepleted embryo), treated with 50 µg/ml tetracycline. Solid lines indicate mean (D) and median (E) values. (TIF)

Figure S7 The effect of oxacillin on zebrafish embryos infected with *S. aureus* strains. (A) Mortality of zebrafish infected with BH1CC *AmecA*::*tetR* (OxS) alone, treated with a range of oxacillin doses (n = 40–60 per treated group, n = 80 untreated). (B) Mortality of zebrafish infected with a 1:1 mixture of BH1CC OxS:OxR, treated with 32 µg/ml oxacillin (n = 65–70 per group). (C) Total terminal CFU load per embryo infected with a 1:1 mixture of BH1CC OxS:OxR, treated with 32 µg/ml oxacillin. (D) Terminal BH1CC OxS:OxR strain ratio per *pu*.1 morphant (phagocyte-depleted embryo), treated with 32 µg/ml oxacillin. (E) Terminal BH1CC OxS:OxR strain ratio in zebrafish treated with 16 µg/ml oxacillin. (F) Mortality of zebrafish infected with either SH1000 or KC043 (*katA ahpC*), treated with a range of oxacillin doses (n = 20–22 per group). Solid lines indicate mean (C) and median (D, E) values.

(TIF)

Figure S8 The effect of 100  $\mu$ g/ml (sub-curing) erythromycin on zebrafish embryos infected with *S. aureus* SH1000 strains. (A) Mortality of zebrafish infected with either SH1000 EryR or TetR individually (n = 80–90 per group). (B) Mortality of zebrafish infected with a 1:1 mixture of SH1000 EryR:TetR (n = 80 per group). (C) Total terminal CFU load per embryo infected with a 1:1 mixture of SH1000 EryR:TetR. Solid lines indicate mean values. (TIF)

Figure S9 The effect of sub-curative antibiotic doses on the pattern of *S. aureus* NewHG infection in different murine organs. EryR/TetR strain ratio is given for (A) kidneys, (B) livers and (C) spleens at two days post infection. Only organs that contained bacterial CFU above the limit of detection are shown. Solid lines indicate median values. (TIF)

## References

- Thwaites GE, Edgeworth JD, Gkrania-Klotsas E, Kirby A, Tilley R, et al. (2011) Clinical management of *Staphylococcus aureus* bacteraemia. Lancet Infect Dis 11: 208–222.
- Cheng AG, DcDent AC, Schneewind O, Missiakas D (2011) A play in four acts: Staphylococcus aureus abscess formation. Trends Microbiol 19: 225–232.
- Horst SA, Hoerr V, Beineke A, Kreis C, Tuchscherr L, et al. (2012) A novel mouse model of *Staphylococcus aureus* chronic osteomyelitis that closely mimics the human infection: an integrated view of disease pathogenesis. Am J Pathol 181: 1206–1214.
- Hume EB, Cole N, Khan S, Garthwaite LL, Aliwarga Y, et al. (2005) A Staphylococcus aureus mouse keratitis topical infection model: cytokine balance in different strains of mice. Immunol Cell Biol 83: 294–300.
- Prajsnar TK, Hamilton R, Garcia-Lara J, McVicker G, Williams A, et al. (2012) A privileged intraphagocyte niche is responsible for disseminated infection of *Staphylococcus aureus* in a zebrafish model. Cell Microbiol 14: 1600–1619.
- Rogers DE, Tompsett R (1952) The survival of staphylococci within human leukocytes. J Exp Med 95: 209–230.
- Thwaites GE, Gant V (2011) Are bloodstream leukocytes Trojan Horses for the metastasis of *Staphylococcus aureus*? Nat Rev Microbiol 9: 215–222.
- Gresham HD, Lowrance JH, Caver TE, Wilson BS, Cheung AL, et al. (2000) Survival of *Staphylococcus aureus* inside neutrophils contributes to infection. J Immunol 164: 3713–3722.
- Grant AJ, Restif O, McKinley TJ, Sheppard M, Maskell DJ, et al. (2008) Modelling within-host spatiotemporal dynamics of invasive bacterial disease. PLoS Biol 6: e74.
- Plaut RD, Kelly VK, Lee GM, Stibitz S, Merkel TJ (2012) Dissemination bottleneck in a murine model of inhalational anthrax. Infect Immun 80: 3189–3193.
- Mertens R, Seyler L, Lacor P, Denis O (2012) Community associated methicillin-resistant Staphylococcus aureus. Acta Clin Belg 67: 235–240.
- Otter JA, French GL (2012) Community-associated meticillin-resistant Staphylococcus aureus: the case for a genotypic definition. J Hosp Infect 81: 143–148.
- Levy SB, FitzGerald GB, Macone AB (1976) Changes in intestinal flora of farm personnel after introduction of a tetracycline-supplemented feed on a farm. N Engl J Med 295: 583–588.
- Rinsky JL, Nadimpalli M, Wing S, Hall D, Baron D, et al. (2013) Livestockassociated methicillin and multidrug resistant *Staphylococcus aureus* is present among industrial, not antibiotic-free livestock operation workers in North Carolina. PLoS One 8: e67641.
- Handwerger S, Tomasz A (1985) Antibiotic tolerance among clinical isolates of bacteria. Annu Rev Pharmacol Toxicol 25: 349–380.
- Maisnier-Patin S, Andersson DI (2004) Adaptation to the deleterious effects of antimicrobial drug resistance mutations by compensatory evolution. Res Microbiol 155: 360–369.
- Andersson DI, Hughes D (2011) Persistence of antibiotic resistance in bacterial populations. FEMS Microbiol Rev 35: 901–911.
- Andersson DI, Levin BR (1999) The biological cost of antibiotic resistance. Curr Opin Microbiol 2: 489–493.
- Massey RC, Buckling A, Peacock SJ (2001) Phenotypic switching of antibiotic resistance circumvents permanent costs in *Staphylococcus aureus*. Curr Biol 11: 1810–1814.
- Trzcinski K, Cooper BS, Hryniewicz W, Dowson CG (2000) Expression of resistance to tetracyclines in strains of methicillin-resistant *Staphylococcus aureus*. J Antimicrob Chemother 45: 763–770.
- Hao H, Dai M, Wang Y, Huang L, Yuan Z (2012) Key genetic elements and regulation systems in methicillin-resistant *Staphylococcus aureus*. Future Microbiol 7: 1315–1329.
- Negri MC, Lipsitch M, Blazquez J, Levin BR, Baquero F (2000) Concentrationdependent selection of small phenotypic differences in TEM beta-lactamasemediated antibiotic resistance. Antimicrob Agents Chemother 44: 2485–2491.

# Table S1 Bacterial strains and plasmids used in this study.

(DOCX)

## Acknowledgments

The authors would like to thank James O'Gara (University College Dublin) and Stephan Heeb (University of Nottingham) for the kind gift of bacterial strains and invaluable discussion and assistance. They also thank Jorge Garcia-Lara for guidance and training and the aquarium staff of the MRC Centre for Developmental and Biomedical Genetics for zebrafish husbandry.

### **Author Contributions**

Conceived and designed the experiments: GM TKP AW SAR SJF. Performed the experiments: GM TKP AW NLW. Analyzed the data: GM MB SJF. Wrote the paper: GM SAR SJF.

- Gullberg E, Cao S, Berg OG, Ilback C, Sandegren L, et al. (2011) Selection of resistant bacteria at very low antibiotic concentrations. PLoS Pathog 7: e1002158.
- Andersson DI, Hughes D (2012) Evolution of antibiotic resistance at non-lethal drug concentrations. Drug Resist Updat 15: 162–172.
- Pengov A, Ceru S (2003) Antimicrobial drug susceptibility of *Staphylococcus aureus* strains isolated from bovine and ovine mammary glands. J Dairy Sci 86: 3157– 3163.
- Harrison EM, Paterson GK, Holden MT, Larsen J, Stegger M, et al. (2013) Whole genome sequencing identifies zoonotic transmission of MRSA isolates with the novel *mecA* homologue *mecC*. EMBO Mol Med 5: 509–515.
- Johnson PJ, Levin BR (2013) Pharmacodynamics, Population Dynamics, and the Evolution of Persistence in *Staphylococcus aureus*. PLoS Genet 9: e1003123.
- 28. Abdel Hady W, Bayer AS, Seidl K, Nast CC, Kiedrowski MR, et al. (2013) Reduced Vancomycin Susceptibility in an *in vitro* Catheter-Related Biofilm Model Correlates with Poor Therapeutic Outcomes in Experimental Endocarditis due to Methicillin-Resistant *Staphylococcus aureus*. Antimicrob Agents Chemother.
- Kjeldgaard J, Cohn MT, Casey PG, Hill C, Ingmer H (2012) Residual antibiotics disrupt meat fermentation and increase risk of infection. MBio 3: e00190-00112.
- Nair CG, Chao C, Ryall B, Williams HD (2013) Sub-lethal concentrations of antibiotics increase mutation frequency in the cystic fibrosis pathogen *Pseudomonas aeruginosa*. Lett Appl Microbiol 56: 149–154.
- Davies JC, Bilton D (2009) Bugs, biofilms, and resistance in cystic fibrosis. Respir Care 54: 628–640.
- Mainiero M, Goerke C, Geiger T, Gonser C, Herbert S, et al. (2010) Differential target gene activation by the *Staphylococcus aureus* two-component system *saeRS*. J Bacteriol 192: 613–623.
- Cheng AG, Kim HK, Burts ML, Krausz T, Schneewind O, et al. (2009) Genetic requirements for *Staphylococcus aureus* abscess formation and persistence in host tissues. FASEB J 23: 3393–3404.
- Barnes PD, Bergman MA, Mecsas J, Isberg RR (2006) *Tersinia pseudotuberculosis* disseminates directly from a replicating bacterial pool in the intestine. J Exp Med 203: 1591–1601.
- Pastar I, Nusbaum AG, Gil J, Patel SB, Chen J, et al. (2013) Interactions of Methicillin Resistant Staphylococcus aureus USA300 and Pseudomonas aeruginosa in Polymicrobial Wound Infection. PLoS One 8: e56846.
- Clatworthy AE, Lee JS, Leibman M, Kostun Z, Davidson AJ, et al. (2009) *Pseudomonas aeruginosa* infection of zebrafish involves both host and pathogen determinants. Infect Immun 77: 1293–1303.
- Brannon MK, Davis JM, Mathias JR, Hall CJ, Emerson JC, et al. (2009) *Pseudomonas aeruginosa* Type III secretion system interacts with phagocytes to modulate systemic infection of zebrafish embryos. Cell Microbiol 11: 755–768.
- Rudkin JK, Edwards AM, Bowden MG, Brown EL, Pozzi C, et al. (2012) Methicillin resistance reduces the virulence of healthcare-associated methicillinresistant *Staphylococcus aureus* by interfering with the *agr* quorum sensing system. J Infect Dis 205: 798–806.
- Kohanski MA, Dwyer DJ, Hayete B, Lawrence CA, Collins JJ (2007) A common mechanism of cellular death induced by bactericidal antibiotics. Cell 130: 797– 810.
- Keren I, Wu Y, Inocencio J, Mulcahy LR, Lewis K (2013) Killing by bactericidal antibiotics does not depend on reactive oxygen species. Science 339: 1213–1216.
- Liu Y, Imlay JA (2013) Cell death from antibiotics without the involvement of reactive oxygen species. Science 339: 1210–1213.
- 42. Cosgrove K, Coutts G, Jonsson IM, Tarkowski A, Kokai-Kun JF, et al. (2007) Catalase (KatA) and alkyl hydroperoxide reductase (AhpC) have compensatory roles in peroxide stress resistance and are required for survival, persistence, and nasal colonization in *Staphylococcus aureus*. J Bacteriol 189: 1025–1035.

- Rauch S, DeDent AC, Kim HK, Bubeck Wardenburg J, Missiakas DM, et al. (2012) Abscess formation and alpha-hemolysin induced toxicity in a mouse model of *Staphylococcus aureus* peritoneal infection. Infect Immun 80: 3721–3732.
- von Kockritz-Blickwede M, Rohde M, Ochmcke S, Miller LS, Cheung AL, et al. (2008) Immunological mechanisms underlying the genetic predisposition to severe *Staphylococcus aureus* infection in the mouse model. Am J Pathol 173: 1657– 1668.
- Schweizer ML, Bossen A, McDanel JS, Dennis LK (2012) Staphylococcus aureus colonization before infection is not associated with mortality among S. aureusinfected patients: a meta-analysis. Infect Control Hosp Epidemiol 33: 796–802.
- Rodriguez CA, Agudelo M, Żuluaga AF, Vesga O (2012) Generic vancomycin enriches resistant subpopulations of *Staphylococcus aureus* after exposure in a neutropenic mouse thigh infection model. Antimicrob Agents Chemother 56: 243–247.
- Bornside GH, Bornside BB, Cohn I, Jr. (1972) Development of tetracyclineresistant Staphylococcus aureus in gnotobiotic mice. Infect Immun 5: 505–512.
- Prajsnar TK, Cunliffe VT, Foster SJ, Renshaw SA (2008) A novel vertebrate model of *Staphylococcus aureus* infection reveals phagocyte-dependent resistance of zebrafish to non-host specialized pathogens. Cell Microbiol 10: 2312–2325.
- Grande Burgos MJ, Kovacs AT, Mironczuk AM, Abriouel H, Galvez A, et al. (2009) Response of *Bacillus cereus* ATCC 14579 to challenges with sublethal concentrations of enterocin AS-48. BMC Microbiol 9: 227.
- 50. Nanduri B, Shack LA, Burgess SC, Lawrence ML (2009) The transcriptional response of *Pasteurella multocida* to three classes of antibiotics. BMC Genomics 10 Suppl 2: S4.
- Petek M, Baebler S, Kuzman D, Rotter A, Podlesek Z, et al. (2010) Revealing fosfomycin primary effect on *Staphylococcus aureus* transcriptome: modulation of cell envelope biosynthesis and phosphoenolpyruvate induced starvation. BMC Microbiol 10: 159.

- Holder D, Berry D, Dai D, Raskin L, Xi C (2013) A dynamic and complex monochloramine stress response in *Escherichia coli* revealed by transcriptome analysis. Water Res 47: 4978–4985.
- Kummerer K, Henninger A (2003) Promoting resistance by the emission of antibiotics from hospitals and households into effluent. Clin Microbiol Infect 9: 1203–1214.
- Chander Y, Kumar K, Goyal SM, Gupta SC (2005) Antibacterial activity of soilbound antibiotics. J Environ Qual 34: 1952–1957.
- Read AF, Day T, Huijben S (2011) The evolution of drug resistance and the curious orthodoxy of aggressive chemotherapy. Proc Natl Acad Sci U S A 108 Suppl 2: 10871–10877.
- Law V, Knox C, Djoumbou Y, Jewison T, Guo AC, et al. (2013) DrugBank 4.0: shedding new light on drug metabolism. Nucleic Acids Res.
- Vagner V, Dervyn E, Ehrlich SD (1998) A vector for systematic gene inactivation in *Bacillus subtilis*. Microbiology 144 (Pt 11): 3097–3104.
- Aish J (2003) Environmental regulation of virulence determinant expression in *Staphylococcus aureus*. Dissertation. University of Sheffield, Sheffield, United Kingdom.
- Cooper EL, Garcia-Lara J, Foster SJ (2009) YsxC, an essential protein in Staphylococcus aureus crucial for ribosome assembly/stability. BMC Microbiol 9: 266.
- Choi KH, Schweizer HP (2006) mini-Tn7 insertion in bacteria with single atfTn7 sites: example Pseudomonas aeruginosa. Nat Protoc 1: 153–161.
- Nüsslein-Volhard C, Dahm R (2002) Zebrafish: a practical approach. New York: Oxford University Press. 303 p.
- Rhodes J, Hagen A, Hsu K, Deng M, Liu TX, et al. (2005) Interplay of Pu.1 and Gata1 determines myelo-erythroid progenitor cell fate in zebrafish. Dev Cell 8: 97–108.